Author's response to reviews

Title: The Karachi Intracranial Stenosis Study (KISS) Protocol: An urban multicenter case-control investigation reporting the clinical, radiologic and biochemical associations of Intracranial Stenosis in Pakistan.

Authors:

Ayeesha Kamran Kamal (ayeesham.kamal@aku.edu)
Fawad Taj (fawadtaj@gmail.com)
Babar Junaidi (babarjunaidi@gmail.com)
Asif Rasheed (dr_asifrasheed@yahoo.com)
Moazzam Zaidi (drmoazzam_2003@yahoo.com)
Muhammed Murtaza (murtaza@consulnet.net)
Naved Iqbal (naveed_iqbl@yahoo.com)
Fahad Hashmat (fahadhashmat@yahoo.com)
Syed Vaqas Alam (vaqasalam@hotmail.com)
Uzma Saleem (dr_uzma_saleem@hotmail.com)
Shahan Waheed (docshahan83@hotmail.com)
Lajpat Bansari (dr_lajpat@yahoo.com)
Nabi Shah (nabi.shah@yahoo.com)
Maria Sameul (marry.samuel@yahoo.com)
Madiha Yameen (madi_ha3@hotmail.com)
Sobia Naz (rameen_ss@yahoo.com)
Farrukh S Khan (fskhan@iname.com)
Naveeduddin Ahmed (dr.sahil03@hotmail.com)
Khalid Mahmood (khalid_khas1@hotmail.com)
Niaz Sheikh (cnmpakistan@gmil.com)
Karim Ullah Makki (asrash03@hotmail.com)
Muhammad M Ahmed (m.masroor@duhs.edu.pk)
Abdul R Memon (asifrasheed2003@gmail.com)
Mohammad Wasay (mohammad.wasay@aku.edu)
Nadir A Syed (nadir.syed@aku.edu)
Bhojo Khealani (bhojo.khealani@aku.edu)
Philippe M Frossard (philippe.frossard@aku.edu)
Danish Salaheen (DS436@medschl.cam.ac.uk)

Version: 2 Date: 9 June 2009

Author's response to reviews: see over
June 5, 2009

To The Editorial Team

BMC Neurology

Dear Editor and Team

We are submitting for review our manuscript entitled “The Karachi Intracranial Stenosis Study (KISS) Protocol: An urban multicenter case-control investigation reporting the clinical, radiologic and biochemical associations of Intracranial Stenosis in Pakistan”.

We believe this protocol has value as it is the first multicentre study of its kind from Pakistan. It shall provide insight into the most common cause of stroke in the developing world populations.

There are no conflicts of interests to declare.

Ayeesha K Kamal, MD, FAHA
Assistant Professor – Neurology
Director Stroke Program, AKUH
Diplomate American Boards Of Psychiatry and Neurology
Diplomate American Board of Vascular Neurology
Consultant Neurologist / Neurointensivist
Aga Khan University Hospital, Stadium Road, PO Box 3500, Karachi-74800 Pakistan.
Email: ayeesha.kamal@aku.edu